Author's response to reviews

Title: Estimation of Cardiac Output and Pulmonary Vascular Resistance by Contrast Echocardiography Transit Time Measurement: a Prospective Pilot Study

Authors:

- Brian G Choi (bchoi@gwu.edu)
- Reza Sanai (reza.sanai@gmail.com)
- Benjamin Yang (benyangmd@gmail.com)
- Heather A Young (youngh@gwu.edu)
- Ramesh Mazhari (rmazhari@mfa.gwu.edu)
- Jonathan S Reiner (jreiner@mfa.gwu.edu)
- Jannet F Lewis (jlewis@mfa.gwu.edu)

Version: 2 Date: 6 August 2014

Author's response to reviews:

As per the e-mail received from Jefferson Victoria on August 6, 2014, the format of the manuscript has been changed as instructed.